%% Antes de processar este arquivo LaTeX (LaTeX2e) deve
%% verificar que o arquivo TEMA.cls estah no mesmo
%% diretorio. O arquivo TEMA.cls pode ser obtido do
%% endereco www.sbmac.org.br/tema.

\documentclass{TEMA}

%\usepackage[brazil]{babel}      % para texto em Português
\usepackage[english]{babel}    % para texto em Inglês

\usepackage[latin1]{inputenc}   % para acentuação em Português
%\input{P-margin.inf}

\usepackage{graphics}
%\usepackage[dvips]{graphics}   % Talvez o correto seja usar este comando e não o de cima
\usepackage{subfigure}
\usepackage{graphicx}
\usepackage{epsfig}
\everymath{\displaystyle}
\usepackage{indentfirst}
\usepackage{here}

\newcommand{\B}{{\tt\symbol{92}}}
\newcommand{\til}{{\tt\symbol{126}}}
\newcommand{\chap}{{\tt\symbol{94}}}
\newcommand{\agud}{{\tt\symbol{13}}}
\newcommand{\crav}{{\tt\symbol{18}}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% Elementos provisórios %%%%%%%%%%%
\usepackage{hyperref}
\hypersetup{
		pdfkeywords={PALAVRAS}{CHAVES}{abnt}{abntex}{abntex2},
	    pdfproducer={LaTeX with abnTeX2}, 	% producer of the document
	    pdfcreator={Julio C Uzinski},
    	colorlinks=true,       		% false: boxed links; true: colored links
    	linkcolor=black,          	% color of internal links
    	citecolor=black,        		% color of links to bibliography
    	filecolor=magenta,      		% color of file links
		urlcolor=blue,
		bookmarksdepth=4
}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{8}
\bibitem{03}
I. Daubechies, ``Ten Lectures on Wavelets''. Philadelphia, PA: SIAM, 1992.

\bibitem{x} T. Froese, S. Hadjiloucas, R.K.H. Galvão, V.M. Becerra, C.J. Coelho, Comparison of extrasystolic ECG signal classifiers using discrete wavelet transforms,\textit{ Pattern Recognit. Lett.}, vol. 27, pp. 393-407, Apr. 2006.

\bibitem{y} R.K.H. Galvão, G.E. José,  H.A.D. Filho, M.C.U. Araujo, E.C. Silva, H.M. Paiva, T.C.B. Saldanha,  E.S.O.N.  Souza, Optimal wavelet filter construction using X and Y data, \textit{Chemometrics and Intelligent Laboratory Systems}, Volume 70, pp. 1-10,  Jan. 2004

\bibitem{05}
S. Mallat, ``A Wavelet Tour of Signal Processing''. San Diego, EUA: Academic Press, 1998.

\bibitem{02} H.M. Paiva, M.N. Martins, R.K.H. Galvao, J. Paiva,  On the space of orthonormal wavelets: additional constraints to ensure two vanishing moments, \textit{Signal Processing Letters}, IEEE , vol.16, no.2, pp.101-104, Feb. 2009.

\bibitem{z}
J. P. L. M. Paiva, C.A. Kelencz, H.M. Paiva, R.K.H. Galvão, M. Magini, Adaptive wavelet EMG compression based on local optimization of filter banks, \textit{Physiological Measurement}, vol. 29, pp. 843-856, 2008.

\bibitem{01} B.G. Sherlock, D.M. Monro, On the space of orthonormal wavelets,\textit{ Signal Processing, IEEE Transactions on}, vol.46, no.6, pp.1716-1720, Jun 1998.

\bibitem{cmacco} J. C. Uzinski, H. M. Paiva, F. Villarreal, M.A.Q. Duarte, R.K.H. Galvão, Additional constraints to ensure three vanishing moments for orthonormal wavelet filter banks, Congresso de matemática aplicada e computacional - Centro-Oeste, Cuiabá - MT, Jul 2013. 

\bibitem{06} P.P. Vaidyanathan, ``Multirate Systems and Filter Banks''. Englewood Cliffs, NJ: Prentice-Hall, 1993.

\bibitem{zou} H. Zou, A.H. Tewfik,  Parametrization of compactly supported orthonormal wavelets, \textit{ Signal Processing, IEEE Transactions on }, vol.41, no.3, pp.1428-1431, Mar 1993.
\end{thebibliography}


%********************************************************
\title
    {Additional constraints to ensure three vanishing moments for orthonormal wavelet filter banks \\ and 
    transient detection%\thanks{The authors would like to thank the Brazilian agency FAPESP (grant 2011/17610-0).} 
     %\thanks{Agradecimentos por auxílio; Trabalho apresentado no Congresso... (se for o caso).}
     }

%\author
%    {J. C. UZINSKI\thanks{juliouzinski@aluno.feis.unesp.br}\,,
%     Department of Electrical Engineering - FEIS,
%     UNESP - Universidade Estadual Paulista, 15385-000 Ilha Solteira, SP, Brazil.
%     \\[1ex] 
%     H. M. PAIVA\thanks{hmpaiva@mectron.com.br}\,,
%     Mectron - Organização Odebrecht,
%     12227-000 São José dos Campos, SP, Brazil.
%     \\[1ex]  
%     F. VILLARREAL\thanks{villa@mat.feis.unesp.br}\,,
%     Department of Mathematics - FEIS, 
%     UNESP - Universidade Estadual Paulista, 15385-000 Ilha Solteira, SP, Brazil.
%     \\[1ex]  
%     M. A. Q. DUARTE\thanks{marco@uems.br}\,,
%     Department of Mathematics - UEMS,
%     Universidade Estadual de Mato Grosso do Sul, 79540-000, Cassilândia, MS, Brazil.
%     \\[1ex]  
%     R. K. H. GALVÃO\thanks{kawakami@ita.br}\,,
%     Department of Electronic Engineering - ITA,
%     12228-900  São José dos Campos, SP, Brazil.}
     
\criartitulo

%\runningheads {Uzinski, Paiva,  Villarreal, Duarte and Kawakami }{Additional constraints to ensure three vanishing moments for orthonormal  wavelet}

\begin{abstract}
{\bf Abstract}. This article presents an improvement to the formulation of Sherlock and Monro for the wavelet parameterization for the obtainment of the restrictions which ensure three vanishing moments. In order to test the formulation presented, a transient signal detection is presented.

{\bf Keywords}. Vanishing Moments, Additional Constraints, Transient detection.
\end{abstract}

%********************************************************
\newsec{Introduction}
Sherlock and Monro \cite{01} started the study of the angular parameterization of orthonormal filter banks, adapting the work of \cite{06} on the factorization of paraunitary matrices and  parameterizing the space of orthonormal wavelets by a set of angular parameters. 

Initially the formulation had a weak point, there were no restrictions to ensure a number of vanishing moments  greater than one. Additional restrictions to ensure at least two vanishing moments were obtained by \cite{02}. This article is an extension of \cite{cmacco} that  presents an improvement to the formulation of \cite{01}, in order to ensure a third vanishing moment for wavelet filter banks, additional constraints are presented to the work of \cite{02} and \cite{01}. An application of this formulation with three vanishing moments for transient detection is presented in this paper. 


Let $$H^{(N)}(z)=\sum_{i=0}^{2N-1}h_i^{(N)}z^{-i}, ~~~ G^{(N)}(z)=\sum_{i=0}^{2N-1}g_i^{(N)}z^{-i}$$ be the transfer functions of the lowpass and highpass filters, respectively, for an orthonormal filter bank with length-$2N$,
%\begin{equation}\label{e01}
%\[H^{(N)}(z)=\sum_{i=0}^{2N-1}h_i^{(N)}z^{-i},\]
%\end{equation}
%\begin{equation}\label{e02}
%\[G^{(N)}(z)=\sum_{i=0}^{2N-1}g_i^{(N)}z^{-i},\]
%\end{equation}
where
\begin{equation}\label{e03}
g_i^{(N)}=(-1)^{i+1}h^{(N)}_{2N-1-i}, \hspace*{1cm} i=0,1,\cdots,2N-1,
\end{equation}
and 
\begin{equation}\label{recursiva}
\begin{array}{ll}
h_0^{(1)}=\cos(\alpha_1)\\
h_1^{(1)}=\sin(\alpha_1)\\
\\
h_0^{(N+1)}=\cos(\alpha_{N+1})h_0^{(N)}\\
h_{2i}^{(N+1)}=\cos(\alpha_{N+1})h_{2i}^{(N)}-\sin(\alpha_{N+1})h_{2i-1}^{(N)}\\
h_{2N}^{(N+1)}=-\sin(\alpha_{N+1})h_{2N-1}^{(N)}\\
\\
h_1^{(N+1)}=\sin(\alpha_{N+1})h_0^{(N)}\\
h_{2i+1}^{(N+1)}=\sin(\alpha_{N+1})h_{2i}^{(N)}+\cos(\alpha_{N+1})h_{2i-1}^{(N)}\\
h_{2N+1}^{(N+1)}=\cos(\alpha_{N+1})h_{2N-1}^{(N)}\\
\end{array} \hspace*{1cm}i=0,1,\cdots,2N-1.
\end{equation}

\newsec{First and second vanishing moments}

If the filter bank is to characterize a wavelet transform, the regularity condition $G^{(N)}(z)\vert_{z=1}=0$  must be satisfied \cite{03,05}. Which, according to \cite{02}, leads to %$$G^{(N)}(z)\vert_{z=1}=\cos\left[\sum_{i=1}^N\alpha_i\right]-\sin\left[\sum_{i=1}^N\alpha_i\right],$$ Devido a condição de regularidade esta equação implica em

\begin{eqnarray}
%\cos\left[\sum_{i=1}^N\alpha_i \right] -\sin\left[\sum_{i=1}^N\alpha_i \right] &=& 0\nonumber\\
%\sqrt{2}\sin\left(\frac{\pi}{4}-\sum_{i=1}^N \right) &=& 0\nonumber\\
%\sin\left(\frac{\pi}{4}-\sum_{i=1}^N \right) &=& 0\nonumber\\
%\frac{\pi}{4}-\sum_{i=1}^N &=& \arcsin(0)\nonumber\\
%\frac{\pi}{4}-\sum_{i=1}^{N-1}-\alpha_N &=& 0\nonumber\\
\alpha_N=\frac{\pi}{4}-\sum_{i=1}^{N-1}\alpha_i.\label{eq01}
\end{eqnarray}
 
According to \cite{05}, to ensure two vanishing moments it is necessary that 
$
\frac{dG^{(N)}(z)}{dz}\bigg\vert_{z=1}=0.
$
This provides, \cite{02},
%\begin{equation}\label{eq03}
%\frac{dG^{(N)}(z)}{dz}\bigg\vert_{z=1}=-N\cos\beta_N+(N-1)\sin\beta_N-\sum_{k=1}^{N-1}(\cos\beta_k+\sin\beta_k),
%\end{equation}
%onde 
%\begin{equation}\label{eq04}
%\left\lbrace \begin{array}{ll}
%\beta_k=\sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i,~ 1\leq k \leq N-1,\\
%\beta_N=\sum_{i=1}^N\alpha_i.
%\end{array}\right.
%\end{equation}
%\par A partir de (\ref{eq01}), (\ref{eq02}) e (\ref{eq03})-(\ref{eq04}) tem-se
%\begin{eqnarray}
%\frac{dG^{(N)}(z)}{dz}\bigg\vert_{z=1} &=& -N \cos\left(\frac{\pi}{4}\right) + N \sin\left(\frac{\pi}{4}\right)-\sin\left(\frac{\pi}{4}\right)-\sum_{i=1}^{N-1}(\cos\beta_k+\sin\beta_k)\nonumber\\
%&=& -\frac{\sqrt{2}}{2}-\sum_{i=1}^{N-1}(\cos\beta_k+\sin\beta_k)\nonumber\\
%&=& -\frac{\sqrt{2}}{2}-\sum_{i=1}^{N-1}\sqrt{2}\sin\left(\beta_k+\frac{\pi}{4}\right)\nonumber\\
%&=& -\frac{\sqrt{2}}{2}-\sum_{i=1}^{N-1}\sqrt{2}\sin\left(\sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i+\sum_{i=1}^N\alpha_i\right)\nonumber\\
%&=&-\frac{\sqrt{2}}{2}-\sqrt{2}\sum_{i=1}^{N-1}\sin\left(\sum_{i=1}^k2\alpha_i\right)\nonumber\\
%&=&\sqrt{2}\left( -\frac{1}{2}-\sum_{i=1}^{N-1}\sin\left(\sum_{i=1}^k2\alpha_i\right)\right)\nonumber%\label{eq05}, 
%\end{eqnarray}
%e de (\ref{eq02}) obtém-se
%que pela condição de regularidade equivale a
%\begin{equation}\label{eq06}
%\sum_{i=1}^{N-1}\left[\sin\sum_{i=1}^k 2\alpha_i\right]=-\frac{1}{2}.
%\end{equation} 

%\par A equação (\ref{eq06}) pode ser escrita em função de $\alpha_{N-1}$
\begin{eqnarray}
%-\frac{1}{2} &=& \sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right]+\sin\sum_{i=1}^{N-1} 2\alpha_i\nonumber\\
%\sin\sum_{i=1}^{N-1} 2\alpha_i &=& -\frac{1}{2}-\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right]\nonumber\\
%\sum_{i=1}^{N-1} 2\alpha_i &=& \arcsin\left\lbrace -\frac{1}{2}-\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right]\right\rbrace \nonumber\\
%\sum_{i=1}^{N-2} 2\alpha_i +2\alpha_{N-1}&=&\arcsin\left\lbrace -\frac{1}{2}-\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right]\right\rbrace \nonumber\\
\alpha_{N-1}&=&\frac{1}{2}\arcsin\left\lbrace -\frac{1}{2}-\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right]\right\rbrace -\sum_{i=1}^{N-2} \alpha_i\label{eq07}.
\end{eqnarray}
Equation (\ref{eq07}) has a real solution if the angles  $\alpha_i$ satisfy the condition 
%$$-1\leq -\frac{1}{2}-\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right] \leq 1$$
%ou equivalente
\begin{equation}\label{eq08}
-\frac{3}{2}\leq\sum_{i=1}^{N-2}\left[\sin\sum_{i=1}^k 2\alpha_i\right] \leq \frac{1}{2}.
\end{equation}
%\par Esta formulação foi proposta por \cite{02} e vai além da proposta de \cite{01} que garantia um momento nulo, pois esta garante o segundo momento nulo. Para além do segundo momento nulo, propomos a seguir a obtenção do terceiro momento nulo. 


\newsec{The Third Vanishing Moment}

In order to obtain a third vanishing moment, \cite{cmacco}, it is necessary that
\begin{equation}\label{eq09}
\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=0.
\end{equation}

Replacing  (\ref{e03}) in the second derivative of $G^{(N)}(z)$ when $z=1$  and writing  conveniently becomes
%\par A derivada segunda de $G^{(N)}(z)$ quando $z=1$ é dada por 
%\begin{equation}\label{eq10}
%\[\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=\sum_{i=1}^{2N-1}g_i^{(N)}(-i)(-i-1),\]
%\end{equation}
%de  $g_i^{(N)}=(-1)^{i+1}h_{2N-1-i}^{(N)}$, \cite{02}, vem que
%\begin{equation}\label{eq11}
%\[\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=\sum_{i=1}^{2N-1}(-1)^{i+1}h_{2N-1-i}^{(N)}(i^2+i),\]
%\end{equation}
%que pode ser escrita como 
%\begin{equation}\label{apd4}
%\[\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=2h^{(N)}_{2N-2}-6h^{(N)}_{2N-3}+\cdots-((2N-2)^2+(2N-2))h^{(N)}_{1}+((2N-1)^2+(2N-1))h^{(N)}_{0},\]
%%\end{equation}
%in an equivalent way
\begin{equation}\label{apd5}
\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=
\begin{array}{ll}
(N^2+2N-1)\sum_{i=0}^{N-1}h_{2i}^{(N)}+\\
\sum_{i=1}^{N-1}(2N+1)\left[\sum_{i=0}^{N-1}h_{2i}^{(N)}-2\sum_{i=N-j+1}^{N-1}h_{2i}^{(N)} \right] +\\
\sum_{j=1}^{N-2}\sum_{k=1}^{N-j-1}2\left[  \sum_{i=0}^{N-1}h_{2i}^{(N)}-2\sum_{i=k+1}^{k+j}h_{2i}^{(N)}\right] -\\
(N^2-1)\sum_{i=0}^{N-1}h_{2i+1}^{(N)}+\\
\sum_{i=1}^{N-1}(2N-1)\left[\sum_{i=0}^{N-1}h_{2i+1}^{(N)}-2\sum_{i=N-j+1}^{N-1}h_{2i+1}^{(N)} \right] -\\
\sum_{j=1}^{N-2}\sum_{k=1}^{N-j-1}2\left[  \sum_{i=0}^{N-1}h_{2i+1}^{(N)}-2\sum_{i=k+1}^{k+j}h_{2i+1}^{(N)}\right].
\end{array}
\end{equation}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{lemmaTEMA}\label{lema1} Considering (\ref{recursiva}), the following equalities are true
\begin{equation}\label{jq1}
\sum_{i=0}^{N-1}h_{2i}^{(N)}=\cos\left[\sum_{i=1}^N\alpha_i \right],
\end{equation} 
 \begin{equation}\label{jq2}
 \sum_{i=0}^{N-1}h_{2i+1}^{(N)}=\sin\left[\sum_{i=1}^N\alpha_i \right].
 \end{equation}
\end{lemmaTEMA}

\begin{proof}Proof by induction:

In the case that $N=1$, to (\ref{jq1}) and (\ref{jq2}), respectively, it has
$$h_0^{(1)}=\cos(\alpha_1) \textrm{ and } h_1^{(1)}=\sin(\alpha_1).$$

Demonstrate that if 
\begin{equation}\label{jq3}
\sum_{i=0}^{N-1}h_{2i}^{(N)}=\cos\left[\sum_{i=1}^N\alpha_i \right],
\end{equation}
 \begin{equation}\label{jq4}
 \sum_{i=0}^{N-1}h_{2i-1}^{(N)}=\sin\left[\sum_{i=1}^N\alpha_i \right]
 \end{equation}
then
\begin{equation}\label{jq5}
\sum_{i=0}^{N+1}h_{2i}^{(N+1)}=\cos\left[\sum_{i=1}^{N+1}\alpha_i \right],
\end{equation}
 \begin{equation}\label{jq6}
 \sum_{i=0}^{N+1}h_{2i+1}^{(N+1)}=\sin\left[\sum_{i=1}^{N+1}\alpha_i \right]
 \end{equation}
\begin{enumerate}
\item[a) ] Demonstrate that the validity of (\ref{jq3}) and (\ref{jq4}) implies the validity of (\ref{jq5}):
\begin{eqnarray}
\sum_{i=0}^{N+1}h_{2i}^{(N+1)} &=& \sum_{i=0}^{N}\left[ \cos(\alpha_{N+1})h_{2i}^{(N)}-\sin(\alpha_{N+1})h_{2i-1}^{(N)}\right] \nonumber\\
&=& \cos(\alpha_{N+1})\sum_{i=0}^{N}\left[h_{2i}^{(N)}\right] -\sin(\alpha_{N+1})\sum_{i=0}^{N}\left[h_{2i-1}^{(N)}\right]\nonumber\\
&=&  \cos(\alpha_{N+1})\cos\left[ \sum_{i=1}^N\alpha_i\right] -\sin(\alpha_{N+1})\sin\left[ \sum_{i=1}^N\alpha_i\right]\nonumber\\
&=& \cos\left[ \alpha_{N+1}+\sum_{i=1}^N\alpha_i\right] =\cos \left[ \sum_{i=1}^{N+1}\alpha_i\right]\nonumber%\label{jq7}
\end{eqnarray}

\item[b) ] Demonstrate that the validity of (\ref{jq3}) and (\ref{jq4}) implies the validity of (\ref{jq6}):
\begin{eqnarray}
\sum_{i=0}^{N+1}h_{2i+1}^{(N+1)} &=& \sum_{i=0}^{N}\left[ \sin(\alpha_{N+1})h_{2i}^{(N)}+\cos(\alpha_{N+1})h_{2i-1}^{(N)}\right] \nonumber\\
&=& \sin(\alpha_{N+1})\sum_{i=0}^{N}\left[h_{2i}^{(N)}\right]+\cos(\alpha_{N+1})\sum_{i=0}^{N}\left[h_{2i-1}^{(N)}\right] \nonumber\\
&=& \sin(\alpha_{N+1})\cos\left[ \sum_{i=1}^N\alpha_i\right]+\cos(\alpha_{N+1})\sin\left[ \sum_{i=1}^N\alpha_i\right]\nonumber\\
&=& \sin \left[(\alpha_{N+1})+\sum_{i=1}^N\alpha_i\right]=\sin\left[ \sum_{i=1}^{N+1}\alpha_i\right]\nonumber%\label{jq8}
\end{eqnarray}
\end{enumerate}
\end{proof}

\begin{lemmaTEMA}\label{lema2}
The equations in (\ref{recursiva}) imply
\begin{equation}\label{jq9}
\sum_{i=1}^{N-1} \left[\sum_{i=0}^kh_{2i}^{(N)}-\sum_{i=0}^{N-1}h_{2i}^{(N)} \right]
= \sum_{i=1}^{N-1}\cos\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i\right] ,
\end{equation}
\begin{equation}\label{jq10}
\sum_{i=1}^{N-1}\left[-\sum_{i=0}^kh_{2+i}^{(N)}+\sum_{i=0}^{N-1}h_{2+i}^{(N)} \right] = \sum_{i=1}^{N-1} \sin\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i\right].
\end{equation}
\end{lemmaTEMA}

\begin{proof}
For $N=1$ the verification of the validity of  (\ref{jq9}) and (\ref{jq10}) is  immediate.

For $N>1$  show that if
\begin{equation}\label{jq11}
\sum_{i=1}^{N-1}\left[ \sum_{i=0}^kh_{2i}^{(N)}-\sum_{i=k+1}^{N-1}h_{2i}^{(N)}\right] = \sum_{i=1}^{N-1}\cos\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i\right] 
\end{equation}
then
\begin{equation}\label{jq12}
\sum_{i=1}^{N+1}\left[ \sum_{i=0}^kh_{2i}^{(N+1)}-\sum_{i=k+1}^{N+1}h_{2i}^{(N+1)}\right] = \sum_{i=1}^{N+1}\cos\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^{N+2}\alpha_i\right],
\end{equation}
and if
\begin{equation}\label{jq13}
\sum_{i=1}^{N-1}\left[-\sum_{i=0}^kh_{2+i}^{(N)}+\sum_{i=0}^{N-1}h_{2+i}^{(N)} \right] = \sum_{i=1}^{N-1} \sin\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^N\alpha_i\right]
\end{equation}
then
\begin{equation}\label{jq14}
\sum_{i=1}^{N+1}\left[-\sum_{i=0}^kh_{2+i}^{(N+1)}+\sum_{i=0}^{N+1}h_{2+i}^{(N+1)} \right] = \sum_{i=1}^{N+1} \sin\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^{N+2}\alpha_i\right].
\end{equation}

\bigskip

\noindent \textbf{ Demonstrate that the validity of  (\ref{jq11})  implies the validity of (\ref{jq12}):}
\begin{eqnarray*}
\sum_{i=1}^{N+1}\left[ \sum_{i=0}^kh_{2i}^{(N+1)}-\sum_{i=k+1}^{N+1}h_{2i}^{(N+1)}\right] &=& 
\sum_{i=1}^{N+1}\left[
\sum_{i=0}^k\left[ \cos(\alpha_{N+1})h_{2i}^{(N)}-\sin(\alpha_{N+1})h_{2i-1}^{(N)}\right]-\right.\\ & & \left. \sum_{i=k+1}^{N+1}\left[ \cos(\alpha_{N+1})h_{2i}^{(N)}-\sin(\alpha_{N+1})h_{2i-1}^{(N)}\right]\right]\\ &=&\sum_{i=1}^{N+1}\left[
\cos(\alpha_{N+1})\sum_{i=0}^kh_{2i}^{(N)}-\sin(\alpha_{N+1})\sum_{i=0}^kh_{2i-1}^{(N)}-\right. \\ & &\left. \cos(\alpha_{N+1})\sum_{i=k+1}^{N+1}
h_{2i}^{(N)}+\sin(\alpha_{N+1})\sum_{i=k+1}^{N+1}h_{2i-1}^{(N)}\right].
\end{eqnarray*}
From Lemma \ref{lema1} it follows that 
\begin{eqnarray*}
\sum_{i=1}^{N+1}\left[ \sum_{i=0}^kh_{2i}^{(N+1)}-\sum_{i=k+1}^{N+1}h_{2i}^{(N+1)}\right] &=& \sum_{i=1}^{N+1}\left[
\cos(\alpha_{N+1})\cos\sum_{i=0}^k\alpha_i-\sin(\alpha_{N+1})\sin\sum_{i=0}^k\alpha_i-\right. \\ & &\left. 
\cos(\alpha_{N+1})\cos\sum_{i=k+1}^{N+1}\alpha_i+\sin(\alpha_{N+1})\sin\sum_{i=k+1}^{N+1}\alpha_i\right] \\&=&
\sum_{i=1}^{N+1}\left[\cos\left(\alpha_{N+1}+\sum_{i=0}^k\alpha_i \right) -\cos\left(\alpha_{N+1}+\sum_{i=k+1}^{N+1}\alpha_i \right)\right]\\&=&
\sum_{i=1}^{N+1}\cos\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^{N+2}\alpha_i\right].
\end{eqnarray*}

\noindent \textbf{Demonstrate that the validity of (\ref{jq13})  implies the validity of (\ref{jq14}):}
\begin{eqnarray*}
\sum_{i=1}^{N-1}\left[-\sum_{i=0}^kh_{2+i}^{(N)}+\sum_{i=0}^{N-1}h_{2+i}^{(N)} \right] &=&\sum_{i=1}^{N-1}\left[
-\sum_{i=1}^k\left[ \sin(\alpha_{N+1})h_{2i}^{(N)}+\cos(\alpha_{N+1})h_{2i-1}^{(N)}\right] +\right. \\ & &\left. 
\sum_{i=k+1}^k\left[ \sin(\alpha_{N+1})h_{2i}^{(N)}+\cos(\alpha_{N+1})h_{2i-1}^{(N)}\right]\right] \\ &=&
\sum_{i=1}^{N-1}\left[
- \sin(\alpha_{N+1})\sum_{i=1}^kh_{2i}^{(N)}-\cos(\alpha_{N+1})\sum_{i=1}^kh_{2i-1}^{(N)} +\right. \\ & &\left. 
 \sin(\alpha_{N+1})\sum_{i=k+1}^kh_{2i}^{(N)}+\cos(\alpha_{N+1})\sum_{i=k+1}^kh_{2i-1}^{(N)}\right].
\end{eqnarray*}
From Lemma \ref{lema1} it follows that 
\begin{eqnarray*}
\sum_{i=1}^{N-1}\left[-\sum_{i=0}^kh_{2+i}^{(N)}+\sum_{i=0}^{N-1}h_{2+i}^{(N)} \right] &=&
\sum_{i=1}^{N-1}\left[-\sin(\alpha_{N+1})\cos\sum_{i=0}^k\alpha_i-\cos(\alpha_{N+1})\sin\sum_{i=0}^k\alpha_i +\right. \\ & &\left. \sin(\alpha_{N+1})\cos\sum_{i=k+1}^{N+1}\alpha_i+\cos(\alpha_{N+1})\sin\sum_{i=k+1}^{N+1}\alpha_i\right]\\ &=&
\sum_{i=1}^{N+1}\left[-\sin\left(\alpha_{N+1}+\sum_{i=1}^k\alpha_i\right)+
                       \sin\left(\alpha_{N+1}+\sum_{i=1}^k\alpha_i\right)    \right]\\ &=&
  \sum_{i=1}^{N+1} \sin\left[ \sum_{i=1}^k\alpha_i-\sum_{i=k+1}^{N+2}\alpha_i\right].                     
\end{eqnarray*}
\end{proof}

From the lemmas \ref{lema1}  and \ref{lema2} it follows that (\ref{apd5}) can be written as
%\begin{equation}\label{apd6}
%\[\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=
%\begin{array}{ll}
%(N^2+2N-1)\cos\left[\sum_{i=1}^N\alpha_i \right]-(N^2-1)\sin\left[\sum_{i=1}^N\alpha_i \right] +\\
%\sum_{i=1}^{N-1}(2N+1)\cos\left[\sum_{i=1}^N\alpha_i-2\sum_{i=(N-j)+1}^N\alpha_i \right] +\\
%\sum_{i=1}^{N-1}(2N-1)\sin\left[\sum_{i=1}^N\alpha_i-2\sum_{i=(N-j)+1}^N\alpha_i \right]+\\
%\sum_{j=1}^{N-2}\sum_{k=1}^{N-j-1}2\cos\left[\sum_{i=1}^N\alpha_i-2\sum_{i=k+1}^{k+j}\alpha_i \right] -\\
%\sum_{j=1}^{N-2}\sum_{k=1}^{N-j-1}2\sin\left[\sum_{i=1}^N\alpha_i-2\sum_{i=k+1}^{k+j}\alpha_i \right],
%\end{array}
%\]%\end{equation}
%or in an equivalent way
\begin{equation}\label{apd7}
\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=
\begin{array}{ll}
(N^2+2N-1)\cos\beta-(N^2-1)\sin\beta+ \\
\displaystyle\sum_{j=1}^{N-1}\bigg\lbrace(2N+1)\cos\beta_j+(2N-1)\sin\beta_j\bigg\rbrace+ \\
\displaystyle\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\left( 2\cos\beta_{j,k}-2\sin\beta_{j,k}\right)   \bigg\rbrace
\end{array}
\end{equation}
where
\begin{equation}\label{apd77}
\left\lbrace \begin{array}{lll}
\beta=\sum_{i=1}^N\alpha_i\\
\beta_j=\sum_{i=1}^N\alpha_i-2\sum_{i=(N-j)+1}^N\alpha_i,\hspace*{1cm} 1\leq j\leq N-1\\
\beta_{j,k}=\sum_{i=1}^N\alpha_i-2\sum_{i=k+1}^{k+j}\alpha_i,\hspace*{1cm} 1\leq j\leq N-1\textrm{ e }1\leq k\leq N-1.
\end{array} \right.
\end{equation}%\]

From equation (\ref{eq01}), equation (\ref{apd77}) has the following implications:  
\[
\begin{array}{lll}
\beta=\frac{\pi}{4}, ~\beta_j=\frac{\pi}{4}-2\sum_{i=(N-j)+1}^N\alpha_i \textrm{ e }
\beta_{j,k}=\frac{\pi}{4}-2\sum_{i=k+1}^{k+j}\alpha_i.
\end{array} 
\]
Leading to some properties:
%\begin{equation}\label{b1}
\[(N^2+2N-1)\cos\beta-(N^2-1)\sin\beta = 2N\cos\frac{\pi}{4},
\]%\end{equation} 
%\begin{equation}\label{b2}
\[\cos\left[\frac{\pi}{4}-2\sum_{i=(N-j)+1}^N\alpha_i\right] +\sin \left[\frac{\pi}{4}-2\sum_{i=(N-j)+1}^N\alpha_i\right]=2\cos\frac{\pi}{4}\cos\left[2\sum_{i=(N-j)+1}^N\alpha_i\right],
\]%\end{equation}
%\begin{equation}\label{b3}
\[\cos\left[\frac{\pi}{4}-2\sum_{i=k+1}^{k+j}\alpha_i\right] -\sin \left[\frac{\pi}{4}-2\sum_{i=k+1}^{k+j}\alpha_i\right]=2\cos\frac{\pi}{4}\sin\left[2\sum_{i=k+1}^{k+j}\alpha_i \right],
\]%\end{equation}
%\begin{equation}\label{b4}
\[\cos\left[\frac{\pi}{4}-2\sum_{i=(N-j)+1}^N\alpha_i\right] =\cos\frac{\pi}{4}\left( \cos\left[2\sum_{i=(N-j)+1}^N\alpha_i\right]+\sin\left[2\sum_{i=(N-j)+1}^N\alpha_i\right]\right) .
\]%\end{equation}

\bigskip

Using these properties, equation (\ref{apd7}-\ref{apd77}) can be written as:
\begin{equation}\label{xomano}
\frac{d^2G^{(N)}(z)}{dz^2}\bigg\vert_{z=1}=
\begin{array}{ll}
2N\cos\frac{\pi}{4}+ 
\sum_{j=1}^{N-1}\bigg\lbrace 2\cos\frac{\pi}{4}(\cos\lambda_j+\sin\lambda_j)+2(2N-1)\cos\frac{\pi}{4}\cos\lambda_j\bigg\rbrace+ \\
\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\left(4\cos\frac{\pi}{4}\sin\lambda_k \right) \bigg\rbrace 
\end{array}
\end{equation}
where
 $\lambda_j=\left[2\sum_{i=(N-j)+1}^N\alpha_i\right]$ e $\lambda_k=\left[ 2\sum_{i=k+1}^{k+j}\alpha_i\right]  $.

\bigskip

Applying (\ref{eq09}) in (\ref{xomano}) 
\begin{eqnarray}
%0 &=& N\frac{\sqrt{2}}{2}+
%\sum_{j=1}^{N-1}\bigg\lbrace  
%\frac{\sqrt{2}}{2}\left[\cos\lambda_j +\sin\lambda_j \right]
%+(2N-1)\frac{\sqrt{2}}{2}\cos\lambda_j\bigg\rbrace+ 
%\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\left(2\frac{\sqrt{2}}{2}\sin\lambda_k \right) \bigg\rbrace \nonumber\\
0 &=& N+\sum_{j=1}^{N-1}\bigg\lbrace  
\sin\lambda_j
+2N\cos\lambda_j\bigg\rbrace+ 
\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\left(2\sin\lambda_k \right) \bigg\rbrace ,\label{emperrada}
\end{eqnarray}
%ou de outra forma 
%\begin{equation}\label{coringa}
%\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k   \bigg\rbrace = -\sum_{j=1}^{N-1}\bigg\lbrace  
%\frac{1}{2}\sin\lambda_j 
%+N\cos\lambda_j  \bigg\rbrace -\frac{N}{2}.
%\end{equation}
Then  (\ref{emperrada}) should be written in terms of  $\alpha_{N-2}$, firstly rewriting (\ref{emperrada}),
\begin{equation}\label{emperrada2}
\sum_{j=1}^{N-1}\bigg\lbrace  \sin\lambda_j +2N\cos\lambda_j \bigg\rbrace+ 
\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}2\sin\left[ \sum_{i=k+1}^{k+j}2\alpha_i\right]    \bigg\rbrace +N=0.
\end{equation}
Decomposing the second parcel of (\ref{emperrada2}) gives
\[
\sum_{j=1}^{N-1}\bigg\lbrace  \sin\lambda_j +2N\cos\lambda_j  \bigg\rbrace +2\sin\left[\sum_2^{N-1}2\alpha_i \right]+  \sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}2\sin\left[ \sum_{i=k+1}^{k+j}2\alpha_i\right]    \bigg\rbrace =-N
\]
and following the reasoning
\begin{eqnarray}
2\sin\left[\sum_2^{N-1}2\alpha_i \right] = -\left\lbrace  \sum_{j=1}^{N-1}\bigg\lbrace  
\sin\lambda_j  
+2N\cos\lambda_j  \bigg\rbrace+ 
\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}2\sin\lambda_k   \bigg\rbrace +N\right\rbrace \nonumber\\
%\sin\left[\sum_2^{N-1}2\alpha_i \right] = -\left\lbrace  \sum_{j=1}^{N-1}\bigg\lbrace  
%\frac{1}{2}\sin\lambda_j  
%+N\cos\lambda_j  \bigg\rbrace+ 
%\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k    \bigg\rbrace +\frac{N}{2}\right\rbrace \\
%\\
\sum_2^{N-1}2\alpha_i  = \arcsin\left\lbrace  -\sum_{j=1}^{N-1}\bigg\lbrace  
\frac{1}{2}\sin\lambda_j  
+N\cos\lambda_j  \bigg\rbrace-
\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k \bigg\rbrace -\frac{N}{2}\right\rbrace\nonumber\\
\sum_{i=2,i\neq N-2}^{N-1}2\alpha_i +2\alpha_{N-2} = \arcsin\left\lbrace  -\sum_{j=1}^{N-1}\bigg\lbrace  
\frac{1}{2}\sin\lambda_j  
+N\cos\lambda_j  \bigg\rbrace-
\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k \bigg\rbrace -\frac{N}{2}\right\rbrace \nonumber\\
%\\
%2\alpha_{N-2} = \arcsin\left\lbrace  -\sum_{j=1}^{N-1}\bigg\lbrace  
%\frac{1}{2}\sin\lambda_j  
%+N\cos\lambda_j  \bigg\rbrace-
%\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k \bigg\rbrace -\frac{N}{2}\right\rbrace -\sum_{i=2,i\neq N-2}^{N-1}2\alpha_i
\alpha_{N-2} = \frac{1}{2}\arcsin\left\lbrace  -\sum_{j=1}^{N-1}\bigg\lbrace  
\frac{1}{2}\sin\lambda_j
+N\cos\lambda_j  \bigg\rbrace-
\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k \bigg\rbrace -\frac{N}{2}\right\rbrace -\sum_{i=2,i\neq N-2}^{N-1}\alpha_i.\nonumber\\\label{an2result}\label{God}
\end{eqnarray}
%
%\par Substituindo (\ref{coringa}) em (\ref{an2result}) tem-se
% 
%
%\begin{eqnarray}
%\alpha_{N-2} &=& \frac{1}{2}\arcsin\left\lbrace  \displaystyle\sum_{j=1}^{N-2}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\left[ \displaystyle\sum_{i=k+1}^{k+j}2\alpha_i\right]  -
%\displaystyle\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\left[ \displaystyle\sum_{i=k+1}^{k+j}2\alpha_i\right]    \bigg\rbrace \right\rbrace -\sum_{i=2,i\neq N-2}^{N-1}\alpha_i\nonumber\\
%\alpha_{N-2} &=& \frac{1}{2}\arcsin(\sin\alpha_2)-\sum_{i=2,i\neq N-2}^{N-1}\alpha_i\nonumber\\
%\alpha_{N-2} &=&  \frac{\alpha_2}{2}-\sum_{i=2,i\neq N-2}^{N-1}\alpha_i. \label{God}
%\end{eqnarray}
Equation (\ref{God}) ensures the third vanishing moment, but the equation  (\ref{God}) has a real solution if the angles  $\alpha_i$ satisfy the condition
\begin{equation}\label{an2re3} -1-\frac{N}{2}\leq\left\lbrace  \sum_{j=1}^{N-1}\bigg\lbrace  
\frac{1}{2}\sin\lambda_j
+N\cos\lambda_j  \bigg\rbrace+
\sum_{j=1}^{N-3}\bigg\lbrace\sum_{k=1}^{N-j-1}\sin\lambda_k \bigg\rbrace \right\rbrace\leq 1-\frac{N}{2}.
\end{equation}
%\end{small}
%lembrando que  $\lambda_j=\left[2\sum_{i=(N-j)+1}^N\alpha_i\right]$ e $\lambda_k=\left[ 2\sum_{i=k+1}^{k+j}\alpha_i\right]  $.
\newsec{Transient detection of a signal}

Consider a orthonormal filter bank with length-8 ($N=4$), which initial configuration characterizes a wavelet with at least one vanishing moment, $\alpha_{p=1}=\{-17.38^{\circ}, 16.83^{\circ}, -45.10^{\circ}, 90.65^{\circ}\}$. To ensure two vanishing moments complies (\ref{eq07}) resulting in  $\alpha_{p=2}=\{-17.38^{\circ}, 16.83^{\circ}, 3.12^{\circ}, 42.43^{\circ}\}$. To ensure at least three va\-nis\-hing moments apply (\ref{God}) which leads to  $\alpha_{p=3}=\{-17.38^{\circ}, -47.23^{\circ}, 93.44^{\circ}, 16.17^{\circ}\}$. Figure \ref{wave} shows the functions $\psi(t)$: $\alpha_{p=1}$, $\alpha_{p=2}$ and $\alpha_{p=3}$, respectively. Each wavelet has the sampling frequency which is denoted by $\omega_s=2\pi/Ts$. 


\begin{figure}[H]
\centering                                          
\epsfig{file=wave.eps,height=9cm,width=13cm} %
\caption{ On the left: $\alpha_{p=1}$, $\alpha_{p=2}$ and $\alpha_{p=3}$. On the right: the sampling frequency of each wavelet.} %
\label{wave}
\end{figure}

Let $f(t)$ be the signal shown in Figure \ref{fig1},
\[
f(t)=
\left\{\begin{array}{ll}
\frac{t}{e^t} ~~~~~~ \textrm{ if }~~  0\leq t < 377\\ 
\frac{(t-0.001)}{e^t} ~~~~~ \textrm{ if }~~   377 \leq t < 995\\
0.3683  ~~~~~~  \textrm{ if }~~  995\leq t \leq 1200.
\end{array}\right.
\]
\begin{figure}[h]
\centering                                          
\hspace*{.25cm}\epsfig{file=11jun004.eps,height=2cm,width=13cm} %
\caption{Signal $f(t)$.} %
\label{fig1}
\end{figure}

This signal has two transients (discontinuities) and was analyzed using $\alpha_{p=1}$, $\alpha_{p=2}$ and $\alpha_{p=3}$, according to figures \ref{dec1}, \ref{dec2} and \ref{dec3}. Each figure shows the decomposition of the first and second wavelet levels, respectively.


\begin{figure}[p]
\centering                                          
\epsfig{file=52.eps,height=5cm,width=12.4cm} %
\caption{Analysis of $f(t)$ with $\alpha_{p=1}$ in the first and second level of decomposition.} %
\label{dec1}
\end{figure}

\begin{figure}[p]
\centering                                          
\epsfig{file=55.eps,height=5cm,width=12.4cm} %
\caption{Analysis of $f(t)$ with $\alpha_{p=2}$ in the first and second level of decomposition.} %
\label{dec2}
\end{figure}

\begin{figure}[p]
\centering                                          
\epsfig{file=58.eps,height=5cm,width=12.4cm} %
\caption{Analysis of $f(t)$ with $\alpha_{p=3}$ in the first and second level of decomposition.} %
\label{dec3}
\end{figure}

%\newpage
Comparing the figures \ref{dec1}, \ref{dec2} and \ref{dec3} it is noticed that despite of the good identification of transients using $\alpha_{p=2}$, the analysis with $\alpha_{p=3}$ also provides a good result, now for $\alpha_{p=1}$ the detection is not quite clear. The amplitude of the detail coefficients, besides the transients that appear in the first decomposition level with $\alpha_{p=2}$, decrease in the case of $\alpha_{p=3}$.The presence of high frequency coefficients indicates that the transients are slightly more highlighted when the analysis is done using $\alpha_{p=3}$.



There are other formulations to work with wavelet filter banks, for example, \cite{zou}, but the formulation of Sherlock and Monro stands for the mathematical and computational simplicities. However, initially there were no constraints to ensure a number of vanishing moments greater than one. An extension of this formulation introducing restrictions to ensure two vanishing moments was done by \cite{02}. In \cite{cmacco}  the constraints to ensure at least three vanishing moments were presented. 

Several papers on applications using this formulation before the extension for three vanishing moments  have been published, some examples are, such as pattern recognition \cite{x}, linear estimation \cite{y}, and signal compression \cite{z}. 


\newsec{Conclusions}



This paper presented the constraints that ensure three vanishing moments  and  also demonstrations and calculations for obtaining  the same. I also presents a brief  application of the formulation for transient detection in signals, a comparative way between wavelets with different regularities. 

In this paper, an application example was used to test the three different wavelets of Sherlock and Monro. Through this example it was noticed that those wavelets are efficient in transient detection, specially when regularity is of at least two vanishing moments. In the case that the parameterization satisfies at least three vanishing moments it was obtained a good identification of transients and better compression or the regular parts of the signal. This fact supports the idea that the more regular is the wavelet the better is the compression of the regular parts of the signal to be decomposed.

\bigskip

\noindent {\Large\textbf{Acknowledgment}}

\noindent The authors would like to thank the Brazilian agency FAPESP (grant 2011/17610-0).




\end{document}
\newpage
$ \  \  $  \thispagestyle{myheadings}  \markboth{      }{   }
